\begin{table}%t1
 %\centering
\par
 \caption{Summary of the properties of EK~Eri.
 \label{tab:results}}
\begin{tabular}{lr@{}ll } \hline\hline
Parameter            &  \multicolumn{2}{c}{Value} & Section \\ \hline
\par
\teff                & $5135$   & $~\pm~80$~K      & \ref{fund_param} \\ 
\logg                & $3.39$   & $~\pm~0.12$      & \ref{fund_param} \\ 
\hspace{0cm}[M/H]    &  $+0.02$ & $~\pm~0.04$      & \ref{fund_param} \\ 
$\xi_t$              & $1.15$   & $~\pm~0.03$~\kms & \ref{fund_param} \\ 
$v_{\rm macro}$    &  $2.3$   & $~\pm~0.2$~\kms  & \ref{period-radius} \\ 
$v\sin i$ (measured) & $<$$1.6$ & $~\pm~0.4$~\kms   & \ref{period-radius} \\ 
$v\sin i$ (predicted)& \multicolumn{2}{l}{$<$0.40~\kms} & \ref{period-radius} \\ \hline 
$L/L_\odot$           & $14.5$   & $~\pm~1.6$      & \ref{mass} \\ 
$M/M_\odot$           & $1.92$   & $~\pm~0.13$     & \ref{mass} \\ 
$R/R_\odot$           & $4.87$   & $~\pm~0.29$     & \ref{mass} \\ 
Age                  &  $1.1$   & $~\pm~0.2$~Gyr  & \ref{mass} \\ \hline 
$P_{\rm rot}$      &  $617.6$ & $~\pm~5.0$~d    & \ref{rv-activity} \\ 
$P_{\rm phot}$     & $308.8$  & $~\pm~2.5$~d    & \ref{period_analysis} \\ 
Ephemeris            &  \multicolumn{2}{l}{HJD$2,453,372.679 + n P_{\rm phot}$} & \ref{period_analysis} \\ 
\par
\hline
\end{tabular}
\end{table}